Author's response to reviews

Title: Efficacy and safety of the human anti-IL-1beta monoclonal antibody canakinumab in rheumatoid arthritis: results of a 12-week, phase II, dose-finding study

Authors:

Rieke Alten (<u>rieke.alten@schlosspark-klinik.de</u>)
Juan Gomez-Reino (<u>juan.gomez-reino.carnota@sergas.es</u>)
Patrick Durez (<u>Patrick.durez@ruma.ucl.ac.be</u>)
Andre Beaulieu (<u>ab5@videotron.ca</u>)
Anthony Sebba (<u>asebba@tampabay.rr.com</u>)
Gerhard Krammer (<u>gerhard.krammer@novartis.com</u>)
Ralph Preiss (<u>ralph.preiss@novartis.com</u>)
Udayasankar Arulmani (<u>udayasankar.arulmani@novartis.com</u>)
Albert Widmer (<u>albert.widmer@novartis.com</u>)
Xavier Gitton (<u>xavier.gitton@novartis.com</u>)
Herbert Kellner (<u>hk@prof-dr-kellner.de</u>)

Version: 4 **Date:** 18 May 2011

Author's response to reviews: see over

Dear Professor John Isaacs

Tel: +44 (0) 20 3192 2013

e-mail: editorial@biomedcentral.com Web: http://www.biomedcentral.com/

We would like to get back to you regarding the final two points raised on the manuscript MS 2063903084916479

Title: Efficacy and safety of the human anti-IL-1beta monoclonal antibody canakinumab in rheumatoid arthritis: results of a 12-week, phase II, dose-finding study

Clarification on no of patient with ALT on page 12:

"In the group receiving canakinumab 300 mg SC q2wk, one patient (1.6%) had an ALT \geq 5 times the ULN and an AST \geq 3 times the ULN, another patient (1.6%) had an ALT \geq 5 times ULN. The latter patient discontinued the study due to the ALT elevation and, for this patient, an elevated ALT of 1.5 times ULN at baseline was noted.

Was changed to:

In the group receiving canakinumab 300 mg SC q2wk, two patients had ALT/ AST elevations: one patient (1.6%) had an ALT \geq 5 times the ULN and an AST \geq 3 times the ULN, another patient (1.6%) had an ALT \geq 5 times ULN. The patient with ALT \geq 5 times ULN discontinued the study due to the ALT elevation, for this patient, an elevated ALT of 1.5 times ULN at baseline was noted.

"A DAS28 of 5.8-5.9 with 11 to 12 swollen joints equates to high disease activity and therefore discussion should be changed"

The sentence "The still-unexplained fact that higher doses of canakinumab did not result in a greater response (dose-response relationship) cannot be accounted for by the relatively low DAS and swollen and tender joint counts "has been deleted from the discussion section of the manuscript following reviewers comments and we have indicated this on page 16 of the manuscript.

We hope with those final edits the manuscript will be acceptable for publication.

With best wishes

Rieke Alten MD,

Department of Internal Medicine II, Rheumatology, Clinical Immunology, Osteology, Schlosspark-Klinik Teaching Hospital of the Charité, University Medicine Berlin, Heubnerweg 2, 14059 Berlin, Germany. *Tel:* +49 30 3264-1325; *Fax:* +49 30 3264-1324;

E-mail: rieke.alten@schlosspark-klinik.de